\begin{table}%t2
\caption{Basic parameters of galaxies in the Canes Venatici I cloud.\label{tab2}}
 \begin{tabular}{llrrrlrrcrrr}
  \hline\hline
\multicolumn{1}{c}{Name}   &
\multicolumn{1}{c}{RA (J2000) Dec} &
\multicolumn{1}{c}{T}&
\multicolumn{1}{c}{$TI$} &
\multicolumn{1}{c}{$V_{\rm LG}$}&
\multicolumn{1}{c}{$D$}&
\multicolumn{1}{c}{$M_B$} &
\multicolumn{1}{c}{log\,$M_{\rm HI}$}&
\multicolumn{1}{c}{log\,$F$}   &
\multicolumn{1}{c}{log\,${\it SFR}$} &
\multicolumn{1}{c}{$p_*$}  &
\multicolumn{1}{c}{$f_*$}    \\
  &  &  &  & &
\multicolumn{1}{c}{Mpc} &
\multicolumn{1}{c}{mag} &
\multicolumn{1}{c}{$M_{\odot}$} &
 &
\multicolumn{1}{c}{$M_{\odot}$\,yr$^{-1}$} &      &       \\
\multicolumn{1}{c}{(1)}&
\multicolumn{1}{c}{(2,~3)}&
\multicolumn{1}{c}{(4)}&
\multicolumn{1}{c}{(5)}&
\multicolumn{1}{c}{(6)}&
\multicolumn{1}{c}{(7)}&
\multicolumn{1}{c}{(8)}&
\multicolumn{1}{c}{(9)}&
\multicolumn{1}{c}{(10)}&
\multicolumn{1}{c}{(11)}&
\multicolumn{1}{c}{(12)}&
\multicolumn{1}{c}{(13)} \\ 
\hline
U5427  &100441.0+292159 & 8& $-$1.2& 424 & 7.1  &$-$14.48&  7.49& $-$13.03& $-$2.19& $ $0.00& $-$0.46 \\
U5672  &102820.9+223417 &10& $-$0.7& 428 & 6.3  &$-$14.65&  7.40& $ $  a  & $-$2.10& $ $0.02& $-$0.64 \\
N3274  &103217.1+274007 & 6& $-$0.3& 461 & 6.5  &$-$16.16&  8.74& $ $  a  & $-$1.11& $ $0.41& $-$0.29 \\
N3344  &104330.2+245525 & 4& $-$1.5& 498 & 6.9  &$-$19.03&  9.67& $-$11.09& $-$0.26& $ $0.11& $-$0.21 \\
U6541  &113329.1+491417 &10& $-$0.7& 304 & 3.89 &$-$13.71&  7.03& $ $  b  & $-$1.68& $ $0.82& $-$1.43 \\
N3738  &113548.6+543122 &10& $-$1.0& 305 & 4.90 &$-$16.61&  8.09& $ $  a  & $-$1.30& $ $0.04& $-$0.75 \\
N3741  &113606.4+451707 &10& $-$0.8& 264 & 3.03 &$-$13.13&  8.05& $ $  a  & $-$2.28& $ $0.45& $ $0.19 \\
KK~109  &114711.2+434019&$-$1& $-$0.6& 241 & 4.51 &$-$9.73&  6.52& $-$15.7:& $-$5.3:& $-$1.2:& $ $1.6: \\
DDO~99  &115053.0+385250 &10& $-$0.5& 248 & 2.64 &$-$13.52&  7.71& $ $  a  & $-$2.48& $ $0.09& $ $0.05 \\
BTS~76  &115844.1+273506 &10& $-$1.2& 451 & 6.3  &$-$12.60&  7.0:& $-$14.17& $-$3.45& $-$0.51& $ $0.3: \\
N4068  &120402.4+523519 &10& $-$1.0& 290 & 4.31 &$-$15.07&  8.13& $ $  a  & $-$1.65& $ $0.30& $-$0.36 \\
MCG~627 &120956.4+362607 &10& $ $0.6& 341 & 4.7  &$-$12.97&  6.7:& $-$14.09& $-$3.62& $-$0.83& $ $0.2: \\
N4144  &120959.3+462726 & 6& $ $0.9& 319 & 7.41 &$-$17.64&  8.79& $ $  a  & $-$0.71& $ $0.21& $-$0.64 \\
N4163  &121208.9+361010 &9& $ $1.4& 164 & 2.96 &$-$13.81&  7.29& $ $  a  & $-$2.83& $-$0.37& $-$0.02 \\
N4190  &121344.6+363760 &10& $ $0.0& 234 & 3.5  &$-$14.33&  7.83& $ $  a  & $-$2.10& $ $0.15& $-$0.21 \\
DDO~113 &121457.9+361308&$-$1& $ $1.6& 283 & 2.86 &$-$11.67&  6.3:& $-$15.0:& $-$5.0:& $-$1.6:& $ $1.1: \\
N4214  &121538.9+361939 & 9& $-$0.7& 295 & 2.94 &$-$17.19&  8.79& $ $  c  & $-$0.64& $ $0.46& $-$0.71 \\
MCG~920 &121546.7+522315 &10& $-$0.7& 245 & 3.4  &$-$12.46&  7.10& $-$13.17& $-$2.98& $ $0.02& $-$0.07 \\
U7298  &121628.6+521338 &10& $-$0.7& 255 & 4.21 &$-$12.27&  7.31& $-$14.49& $-$4.12& $-$1.05& $ $1.29 \\
N4244  &121729.9+374827 & 6& $-$0.0& 255 & 4.49 &$-$18.60&  9.30& $ $  d  & $-$0.90& $-$0.36& $ $0.06 \\
N4258  &121857.5+471814 & 4& $-$0.7& 507 & 7.83 &$-$21.25&  9.81& $ $  e  & $ $0.35& $-$0.17& $-$0.68 \\
U7356  &121909.1+470523 &10& $ $0.4& 330 & 7.19 &$-$13.61&  8.7:& $-$14.64& $-$3.81& $-$1.27& $ $2.4: \\
I3308  &122517.9+264253 & 7& $-$2.:& 277 &12.8  &$-$15.55&  8.68& $-$13.10& $-$1.68& $ $0.08& $ $0.17 \\
KK~144  &122527.9+282857 &10& $-$0.9& 453 & 6.3  &$-$12.59&  7.90& $-$13.83& $-$3.11& $-$0.17& $ $0.87 \\
N4395  &122549.8+333246 & 8& $ $0.1& 315 & 4.67 &$-$17.88&  9.22& $-$11.04& $-$0.56& $ $0.27& $-$0.36 \\
UA~281  &122616.0+482931 & 9& $-$0.8& 349 & 5.43 &$-$13.86&  7.84& $-$11.80& $-$1.21& $ $1.22& $-$1.09 \\
DDO~126 &122705.1+370833 &10& $ $0.1& 231 & 4.87 &$-$14.38&  8.20& $ $  a  & $-$1.91& $ $0.32& $-$0.03 \\
DDO~125 &122741.8+432938 &10& $-$0.9& 240 & 2.54 &$-$14.16&  7.52& $-$12.67& $-$2.74& $-$0.42& $ $0.12 \\
U7584  &122802.9+223522 & 9& $-$0.1& 545 & 7.6  &$-$13.30&  7.59& $-$13.39& $-$2.50& $ $0.16& $-$0.05 \\
KKH~80  &122805.4+221727 &10& $-$0.1& 542 & 7.5  &$-$12.47&  7.02& $-$15.4:& $-$4.5:& $-$1.5:& $ $1.4: \\
N4449  &122811.2+440540 & 9& $-$0.0& 249 & 4.21 &$-$18.27&  9.52& $-$10.43& $-$0.03& $ $0.64& $-$0.59 \\
DDO~127 &122828.5+371400 & 9& $-$0.7& 291 & 6.9  &$-$14.28&  8.15& $-$13.64& $-$2.84& $-$0.57& $ $0.85 \\
U7605  &122839.0+354305 &10& $ $0.7& 317 & 4.43 &$-$13.53&  7.42& $-$12.90& $-$2.49& $ $0.08& $-$0.23 \\
N4460  &122845.8+445152 & 1& $-$0.7& 542 & 9.59 &$-$17.89&  7.65& $-$11.72& $-$0.59& $ $0.23& $-$1.90 \\
KK~149  &122852.3+421040 &10& $-$0.8& 446 & 6.2  &$-$14.06&  7.43& $-$13.39& $-$2.68& $-$0.32& $-$0.03 \\
U7639  &122953.0+473148 &10& $ $0.4& 446 & 7.1  &$-$15.18&  7.79& $-$13.64& $-$2.82& $-$0.91& $ $0.47 \\
KK~151  &123023.8+425405 & 9& $-$0.4& 479 & 6.7  &$-$13.41&  7.32& $-$13.60& $-$2.83& $-$0.21& $ $0.01 \\
DDO~133 &123253.0+313221 &10& $-$1.1& 321 & 6.1  &$-$15.47&  8.58& $ $  c  & $-$1.53& $ $0.26& $-$0.03 \\
Arp~211 &123721.3+384443 &10& $-$0.7& 484 & 6.70 &$-$13.47&  7.13& $ $  b  & $-$2.46& $ $0.13& $-$0.55 \\
UA~292  &123840.0+324600 &10& $-$0.3& 306 & 5.0  &$-$12.47&  7.93& $ $  f  & $-$2.19& $ $0.80& $-$0.02 \\
N4627  &124159.7+323425&$-$3& $-$1.4:&541 & 9.38 &$-$16.87&  7.6:& $-$15.0:& $-$3.9:& $-$2.7:& $ $1.4: \\
N4631  &124208.0+323229 & 7& $-$1.1:&605 & 7.66 &$-$19.74& 10.11& $-$11.04& $ $0.37& $ $0.45& $-$0.40 \\
I3687  &124215.1+383007 &10& $ $1.1& 385 & 4.57 &$-$14.64&  8.03& $ $  a  & $-$2.05& $ $0.07& $-$0.06 \\
KK~160  &124357.4+433941 &10& $ $1.0& 346 & 4.8  &$-$11.52&  6.58& $-$15.1:& $-$4.6:& $-$1.3:& $ $1.1: \\
DDO~147 &124659.8+362835 &10& $-$1.4& 351 & 9.9  &$-$14.94&  8.62& $-$13.26& $-$2.15& $-$0.15& $ $0.63 \\
KK~166  &124913.3+353645&$-$3& $ $0.3&     & 4.74 &$-$10.82&  6.4:& $-$15.15:&$-$4.7:& $-$1.0:& $ $0.9: \\
U7990  &125027.0+282107 &10& $-$2.:& 495 &20.9  &$-$15.46&  8.38& $-$14.01& $-$2.25& $-$0.45& $ $0.49 \\
N4736  &125053.5+410710 & 2& $-$0.5& 353 & 4.66 &$-$19.83&  8.68& $ $  e  & $-$0.19& $-$0.14& $-$1.27 \\
DDO~154 &125405.2+270855 &10& $-$1.1& 355 & 3.91 &$-$13.87&  8.72& $ $  a  & $-$2.08& $ $0.35& $ $0.66 \\
N4826  &125644.2+214105 & 2& $-$0.6& 364 & 4.44 &$-$19.77&  8.42& $ $  g  & $-$0.54& $-$0.47& $-$1.18 \\
I4182  &130549.3+373621 & 9& $ $0.6& 356 & 4.70 &$-$16.40&  8.53& $ $  a  & $-$1.61& $-$0.19& $ $0.00 \\
U8215  &130803.6+464941 &10& $-$0.3& 297 & 4.55 &$-$12.30&  7.32& $-$13.77& $-$3.34& $-$0.28& $ $0.52 \\
N5023  &131211.9+440219 & 6& $-$1.6& 476 & 6.61 &$-$17.12&  8.70& $-$12.03& $-$1.11& $ $0.02& $-$0.33 \\
DDO~167 &131322.8+461911 &10& $ $0.0& 243 & 4.19 &$-$12.70&  7.27& $ $  c  & $-$2.64& $ $0.26& $-$0.23 \\
DDO~168 &131428.6+455510 &10& $-$0.0& 273 & 4.33 &$-$15.28&  8.42& $ $  a  & $-$1.83& $ $0.04& $ $0.11 \\
\hline
DDO\,169 &131530.7+472947 &10& $-$0.2& 345 & 4.23 &$-$13.71&  7.81& $ $  a  & $-$2.40& $ $0.10& $ $0.07 \\
N5204  &132936.4+582504 & 9& $-$1.1& 341 & 4.65 &$-$16.75&  8.76& $ $  a  & $-$1.04& $ $0.24& $-$0.34 \\
N5194  &132956.0+471404 & 5& $ $4.1& 555 & 8.0  &$-$21.34&  9.52& $ $  g  & $ $0.56& $ $0.00& $-$1.18 \\
N5195  &132958.7+471605&$-$1& $ $5.0& 558 & 8.02 &$-$19.22&  8.18& $-$11.33& $-$0.38& $-$0.09& $-$1.58 \\
U8508  &133044.4+545436 &10& $-$1.0& 186 & 2.56 &$-$12.98&  7.34& $-$12.51& $-$2.57& $ $0.22& $-$0.23 \\
N5229  &133402.9+475455 & 7& $-$0.6& 460 & 5.1  &$-$14.60&  8.18& $-$12.46& $-$1.86& $ $0.28& $-$0.10 \\
N5238  &133442.7+513650 & 9& $-$0.9& 345 & 4.24 &$-$14.64&  7.27& $ $  a  & $-$2.04& $ $0.08& $-$0.83 \\
U8638  &133919.4+244633 & 9& $-$1.3& 273 & 4.27 &$-$13.77&  7.22& $-$12.68& $-$2.30& $ $0.17& $-$0.62 \\
DDO~181 &133953.8+404421 &10& $-$1.3& 272 & 3.02 &$-$12.97&  7.39& $-$12.58& $-$2.51& $ $0.28& $-$0.24 \\
DDO~183 &135051.1+380116 &10& $-$1.2& 257 & 3.18 &$-$13.13&  7.36& $ $  a  & $-$3.04& $-$0.31& $ $0.26 \\
HolmIV &135445.1+535417 &10& $ $0.6& 276 & 6.83 &$-$15.48&  8.34& $-$12.40& $-$1.61& $ $0.18& $-$0.19 \\
U8833  &135448.7+355015 &10& $-$1.4& 285 & 3.12 &$-$12.37&  7.14& $ $  e  & $-$3.06& $-$0.03& $ $0.06 \\
U8882  &135714.6+540603&$-$1& $ $0.0&     & 8.3  &$-$13.64&  7.2:& $-$15.04:&$-$4.1:& $-$1.6:& $ $1.2: \\
M101   &140312.8+542102 & 6& $ $0.6& 379 & 7.38 &$-$21.23& 10.36& $ $  e  & $ $0.74& $ $0.23& $-$0.52 \\
N5474  &140502.1+533947 & 8& $ $2.0& 413 & 7.2  &$-$17.74&  9.12& $ $  a  & $-$0.82& $ $0.06& $-$0.20 \\
N5477  &140533.1+542739 & 9& $ $1.4& 443 & 7.7  &$-$15.29&  8.26& $ $  a  & $-$1.27& $ $0.59& $-$0.61 \\
KK~230  &140710.7+350337 &10& $-$1.0& 126 & 1.92 &$-$9.57&  6.35& $-$15.32:&$-$5.6:& $-$1.5:& $ $1.85: \\
KKH~87  &141509.4+570515 &10& $ $1.0& 473 & 7.4  &$-$13.30&  7.64& $-$13.42& $-$2.57& $ $0.09& $ $0.07 \\
DDO~187 &141556.5+230319 &10& $-$1.2& 172 & 2.28 &$-$12.51&  7.16& $ $  a  & $-$3.57& $-$0.59& $ $0.59 \\
N5585  &141948.3+564349 & 4& $-$0.8& 459 & 5.7  &$-$17.82&  8.98& $ $  a  & $-$0.85& $ $0.00& $-$0.31 \\
DDO~190 &142443.5+443133 &10& $-$1.3& 263 & 2.79 &$-$14.18&  7.63& $-$12.53& $-$2.52& $-$0.21& $ $0.01 \\
DDO~194 &143524.6+571524 &10& $-$0.0& 385 & 8.0  &$-$15.05&  8.00& $ $  a  & $-$2.18& $-$0.22& $ $0.04 \\
KKR~25  &161347.6+542216&$-$1& $-$0.7&  68:& 1.86 &$-$9.94&  4.9:& $-$14.64:&$-$5.0:& $-$1.0:& $-$0.2: \\
\hline
\end{tabular}
\smallskip
Notes.\\
a $-$  James et~al. (\cite{James04});
b $-$ Gil de Paz et~al.  (\cite{GildePaz03});
c $-$ Hunter \& Elmegreen (\cite{Hunter04});
d $-$ Strickland et~al. (\cite{Strickland04});
e $-$ Kennicutt et~al. (\cite{Kennicutt89});
f~$-$~van Zee (\cite{vanZee00});
g $-$ Young et~al. (\cite{Young96}).         
\end{table}